

## G OPEN ACCESS

**Citation:** Bezemer JM, Merckx J, Freire Paspuel BP, Calvopiña M, de Vries HJC, Schallig HDFH, et al. (2023) Diagnostic accuracy of qPCR and microscopy for cutaneous leishmaniasis in rural Ecuador: A Bayesian latent class analysis. PLoS Negl Trop Dis 17(11): e0011745. https://doi.org/ 10.1371/journal.pntd.0011745

Editor: Karina Andrea Gómez, INGEBI, ARGENTINA

Received: April 14, 2023

Accepted: October 20, 2023

Published: November 29, 2023

**Copyright:** © 2023 Bezemer et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All data generated or analyzed during this study are included in this published article and its <u>supplementary information</u> files.

**Funding:** Foundation Latin Link Nederland http:// www.latinlink-nederland.nl/ provided funding for the current study (to JMB). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. RESEARCH ARTICLE

## Diagnostic accuracy of qPCR and microscopy for cutaneous leishmaniasis in rural Ecuador: A Bayesian latent class analysis

Jacob M. Bezemer<sup>1,2,3</sup>\*, Joanna Merckx<sup>4,5</sup>, Byron P. Freire Paspuel<sup>6,7</sup>, Manuel Calvopiña<sup>8</sup>, Henry J. C. de Vries<sup>3,9</sup>, Henk D. F. H. Schallig<sup>2,3</sup>, Mariska M. G. Leeflang<sup>5</sup>, Nandini Dendukuri<sup>10</sup>

 Hospital Shell, Fundación Misión Cristiana de Salud, Shell, Pastaza, Ecuador, 2 Department of Medical Microbiology and Infection Prevention, Laboratory for Experimental Parasitology, Amsterdam University Medical Centers location Academic Medical Center at the University of Amsterdam, Amsterdam, the Netherlands, 3 Amsterdam Institute for infection and Immunity, Infectious Diseases Program, Amsterdam, the Netherlands, 4 Department of Epidemiology, Biostatistics, and Occupational Health, McGill University, Montreal, Canada, 5 Department of Epidemiology and Data Science, Amsterdam UMC location University of Amsterdam, Amsterdam, the Netherlands, 6 Laboratorios de Investigación, Universidad de las Américas, Quito, Ecuador, 7 Vall d'Hebron Research Institute, Hospital Universitari Vall d'Hebron, Barcelona, Spain, 8 OneHealth Research Group, Facultad de Medicina, Universidad de las Américas, Quito, Ecuador,
 9 Department of Infectious Diseases, Center for Sexual Health, Public Health Service, Amsterdam, the Netherlands, 10 Research Institute of the McGill University Health Centre, Montreal, Quebec, Canada

These authors contributed equally to this work.
\* j.m.bezemer@amsterdamumc.nl

### Abstract

#### Background

Clinical and laboratory diagnosis of cutaneous leishmaniasis (CL) is hampered by underascertainment of direct microscopy.

#### Methods

This study compared the diagnostic accuracy of qPCR on DNA extracted from filter paper to the accuracy of direct smear slide microscopy in participants presenting with a cutaneous lesion suspected of leishmaniasis to 16 rural healthcare centers in the Ecuadorian Amazon and Pacific regions, from January 2019 to June 2021. We used Bayesian latent class analysis to estimate test sensitivity, specificity, likelihood ratios (LR), and predictive values (PV) with their 95% credible intervals (95%Crl). The impact of sociodemographic and clinical characteristics on predictive values was assessed as a secondary objective.

#### Results

Of 320 initially included participants, paired valid test results were available and included in the diagnostic accuracy analysis for 129 from the Amazon and 185 from the Pacific region. We estimated sensitivity of 68% (95%Crl 49% to 82%) and 73% (95%Crl 73% to 83%) for qPCR, and 51% (95%Crl 36% to 66%) and 76% (95%Crl 65% to 86%) for microscopy in the Amazon and Pacific region, respectively. In the Amazon, with an estimated disease

**Competing interests:** I have read the journal's policy and the authors of this manuscript have the following competing interests: JM was employed by bioMérieux Canada, Inc until July 2021.

prevalence among participants of 73%, negative PV for qPCR was 54% (95%Crl 5% to 77%) and 44% (95%Crl 4% to 65%) for microscopy. In the Pacific, (prevalence 88%) the negative PV was 34% (95%Crl 3% to 58%) and 37% (95%Crl 3% to 63%). The addition of qPCR parallel to microscopy in the Amazon increases the observed prevalence from 38% to 64% (+26 (95%Crl 19 to 34) percentage points).

#### Conclusion

The accuracy of either qPCR on DNA extracted from filter paper or microscopy for CL diagnosis as a stand-alone test seems to be unsatisfactory and region-dependent. We recommend further studies to confirm the clinically relevant increment found in the diagnostic yield due to the addition of qPCR.

#### Author summary

Cutaneous leishmaniasis is caused by the parasite *Leishmania* and is treated when a microscopy test confirms the presence of the parasite in a sample of the lesion. This test, however, is known to miss patients with cutaneous leishmaniasis. DNA diagnostic tests (like PCR) that detect the parasite's genetic material in the lesion, have been proposed to improve diagnosis. Filter paper preserves DNA at room temperature and allows samples to be transported from remote health centers to the PCR laboratory. The ability of microscopic and DNA testing to recognize leishmaniasis patients in real-life is complex to evaluate. We compared the performance of both tests using a statistical method that can evaluate both tests simultaneously without assuming that either test works perfectly. We found that PCR will be positive 68% of the time in a participant with leishmaniasis in the Amazon and 73% of the time in the Pacific region. In the Amazon, microscopy detects one out of every two cases, while it does in three out of every four cases in the Pacific. The addition of the PCR test can improve the number of participants with a diagnosis of leishmaniasis, mostly in the Amazon region.

#### Introduction

#### Background

Protozoan parasites of the genus *Leishmania* are the causative agent of cutaneous, mucosal, and visceral human leishmaniasis. The mainstay method for leishmaniasis confirmation is the combination of clinical characteristics and microscopy [1,2]. Worldwide, cutaneous leishmaniasis (CL) is the most common clinical manifestation of leishmaniasis affecting 600.000 to 1 million people annually [3]. CL manifests mainly as localized skin ulcers and nodules. In Ecuador it has an estimated prevalence of 3905–6415 cases or 30–49 per 100.000 inhabitants per year. It leads to an estimated health loss of 0.32 Disability Adjusted Live Years (DALY) per 100.000 people per year in the country and affects poor and indigenous populations disproportionally [4, 5].

According to the Ecuadorian Ministry of Health (MoH), direct microscopy observation of the parasite (smear or biopsy) is the "gold standard" to diagnose CL in Ecuador. Culture, sero-logical and molecular tests are not available at public health centres as in most endemic regions [6]. The diagnostic accuracy of smear slide microscopy provided in Ecuador has however not

been systematically evaluated. The existing small, limited number of peer-reviewed studies (of 14 to 90 participants) show low sensitivities ranging from 14% to 51% and specificities of near 100%, however based on a comparison to composite reference standards [7–11] which might give a biased result. Patients are nevertheless provided anti-leishmanial treatment for free by the MoH, conditional on having received a positive microscopy diagnosis [12]. Given the aforementioned estimated low sensitivity of the diagnostic test, this might leave several thousands of patients without treatment every year.

Molecular methods are promising for CL diagnosis because of their reported high diagnostic accuracy compared to microscopy [13]. Centralization of molecular tests would save costs, but transportation generally requires a cold chain that is not available. Hashiguchi *et al.* proposed filter paper imprints of CL lesions as a solution for the transport challenges [12]. Filter paper allows prolonged sample DNA conservation without the need for a cold chain [14]. An alternative approach to improve the diagnostic yield is the use of epidemiological, demographic, and clinical characteristics in the diagnostic process [6,15]. Weigle *et al.* reported high sensitivity of the use of a clinical diagnostic algorithm for Colombian CL, but the estimated specificity was low. A corresponding algorithm, however, has to be adapted and validated regionally before its implementation [16]. Of special clinical interest is the predictive value of the tests in use to evaluate the likelihood the participant has or does not have the disease after including the test result [17].

The lack of a reliable reference standard test or thus the absence of a true "gold standard" for CL diagnosis makes it challenging to assess the accuracy of any new diagnostic method and to investigate the true accuracy of the tests in use [18]. Two commonly proposed solutions to this problem are the use of 1) a composite reference standard, and 2) Bayesian Latent Class Analysis (LCA). Diagnostic test accuracy estimates based on a composite reference standard may be biased because combining multiple imperfect tests does not make one perfect reference standard [19], nor can the accuracy of the tests in the composite be evaluated. LCA, in comparison, applies a statistical model that simultaneously analyzes the results of the different tests observed while taking the imperfect nature of each test into account. Provided the assumptions of the model are correct, LCA allows unbiased estimation of the test accuracies, without depending on a perfect reference test [20,21].

In this prospective cross-sectional study, we used Bayesian LCA to estimate the diagnostic accuracy (sensitivity and specificity) of qPCR on DNA extracted from filter paper and microscopy for the diagnosis of CL in Ecuador. As a secondary objective, we assessed the predictive values of specific demographic and clinical criteria in our study population. These objectives are in accordance with the WHO Road Map for Neglected Tropical Diseases, 2021–2030 [22].

#### Methods

#### **Ethics statement**

This prospective study was approved by the ethical committee of the 'Universidad Internacional del Ecuador' (registration number: UIDE-FCM-EDM-COM-18-0069) and by the Ecuadorian MoH (registration number: MSPCURI000284-3). All participants signed a written consent and received free treatment for leishmaniasis according to the Ecuadorian MoH guidelines. The full study protocol can be accessed upon reasonable request to the corresponding author.

#### Participants, data source, and data collection

This is a cross-sectional study of participants suspected of CL. Any case with a suspected cutaneous lesion for whom a physician practicing at a participating health center ordered CL testing was eligible for inclusion. Participants were excluded from the study if they did not provide written informed consent or had no cutaneous lesions. Participants were excluded from the diagnostic accuracy assessment if a qPCR and/or microscopy result was missing. Participants were included at three public primary health care centers in the Pacific subtropical region of the Pichincha province and from public and private primary health care centers and hospitals in the Napo, Pastaza, and Morona Santiago provinces in the Amazon (Total N = 16, See <u>S1 Fig</u>). Participants were identified and enrolled consecutively between the 1<sup>st</sup> of January 2019 and the 30<sup>th</sup> of June 2021 at the centers, and during community visits, by the physician, nurse, or laboratory technician, just before routine sampling for CL. The convenience sampling technique was used to obtain the study population sample size. A separate publication on this data provides additional details on the sample size calculation [23]. All centers had the laboratory capacity to perform smear slide microscopy and offered free treatment for CL-confirmed participants (intramuscular meglumine antimoniate for 20 consecutive days) or treatment for alternative diagnoses. The demographics of participants and clinical characteristics of cutaneous lesions were recorded before sampling. Age was recorded in years and ethnicity was based on participant self-reporting. The variable ethnicity was dichotomized for the analysis in: Amerindian (Amazon Kichwa, Shuar, Achuar, Shiwiar, Zapara, Andwa, or Waorani) and Mestizo (other than Amerindian). Gender was recorded as binary (male/female) and health-seeking delay was defined as the time since lesion onset as mentioned by the participant. The lesion type was classified by the professional into the categories i) ulcer, ii) nodule, and iii) other and presented in the manuscript as proportions of the total number of lesion types (total number of participants with known lesion type plus number of participants with mixed lesion types). The number of lesions was counted by the professional as the number of lesions separated by healthy skin. Body location was drawn on a person figure by the health professional. Body location of the lesion was divided into the categories 'Head and neck', 'Trunk', 'Upper limbs', and 'Lower limbs' for analysis. This variable was presented in the manuscript as proportion of the total number of body locations with lesions (total number of participants with known body location plus number of participants with lesions in multiple body locations). The geographical location of the participant was recorded to estimate the altitude of the place of infection in meters from the altitude of the airstrip of the nearest village (http:// www.ais.aviacioncivil.gob.ec/) or with topographic-map.com (https://es-ec.topographic-map. com/maps/60gw/Ecuador/) and was dichotomized as well in the Amazon and the Pacific region. Species determination was done by sequencing Cytochrome B and MPI as described elsewhere [23]. All the study data were collected on paper forms and entered into an electronic data capture system (https://www.castoredc.com/). Data entry was done in duplicate by two independent investigators and computer validated. We used the STARD-BLCM guidelines for the reporting of the study [24].

#### Sample collection and diagnostic tests

The sample for microscopy was collected by scraping the inner border of the cutaneous lesion. If a participant had more than one lesion, the most recent one was sampled. If a participant presented mixed lesion types (i.e., ulcer and nodule) the ulcer was sampled. The resulting material was collocated on a glass slide and Giemsa stained following the Ecuadorian MoH guidelines [25]. Smear slide microscopy was performed by WHO-trained and experienced microscopists at the local center and sequentially at the central facility for all primary positive smears for diagnosis confirmation. The final results are reported as positive or negative (binary) because Ecuadorian laboratory technicians are not trained in grading parasite density [25]. A positive microscopy test meant it had been read positive on two occasions (on-site and

at the central laboratory). Discordant results (positive on only one occasion) were regarded as negative. Technicians were aware of the clinical characteristics of the participants but unaware of the results of qPCR testing. Laboratory technicians were responsible for reporting adverse reactions during or after sampling to the principal investigator. To collect the sample for qPCR, the local laboratory technician pressed a filter paper (903 Protein Saver Card (Whatman, Newton Center, MA)) three times, for at least one second each, on the inner border of the suspected lesion, immediately after scraping for the smear slide. Filter papers were dried and sent under uncontrolled conditions to the research laboratory of the 'Universidad de las Americas' in Quito by canoe, plane, bus, private car, and/or postal service. They were stored at room temperature in the including centers, during transport, and in the research laboratory in Quito. DNA was extracted from the filter paper according to a Chelex and Proteinase K based protocol [26] that is non-time consuming and cheap and therefore preferable for resourcerestricted settings [27,28]. A piece of 2\*2mm with visible material was separated from the filter paper and placed in a sterile 1.5mL tube containing 200µL of 10% (wt/vol) Chelex 100 (Sigma-Aldrich, USA) and  $10\mu$ L ( $\geq$ 4 units) of Proteinase K (Invitrogen, USA). Samples were vortexed for 5 min, incubated at 56°C for 60 min, and then at 96°C for 20 min. Finally, samples were centrifuged at 10,000 g (earth's gravitational force) for 5 min, and the supernatant containing extracted DNA (approximately  $150\mu L$ ) was removed to another sterile tube. Extracted DNA was quantified with the NanoDrop 2000 Spectrophotometer (Thermo Scientific, USA). Leishmania DNA was detected by duplex real-time PCR (qPCR) of the Leishmania 18S ribosomal (rDNA) gene, which was validated for South American Leishmania species [29] and the human Tumor Necrosis Factor (hTNF) gene as internal control [30]. Leishmania rDNA and hTNF probes were labeled with FAM and HEX as reporters, respectively. The qPCR reaction was prepared with 2µL of extracted DNA and 13µL of master mix containing: 1X TaqMan Universal PCR Master Mix (Applied Biosystems, USA), 500nM of each primer, 250nM of each probe, and nuclease-free water to complete a final volume of 15µL. The reaction was run in a CFX96 Dx Thermal Cycler (Bio-Rad, USA) following this protocol: 95°C for 10 min; 45 cycles of 95°C for 15 sec and 58°C for 60 sec (detection). The last step was optimized with an alignment/extension temperature gradient maintaining the temperature with the highest fluorescence and lowest Ct values. Other conditions were according to the manufacturer's recommendations. The detection limit of the qPCR reaction was determined directly on a 10-fold dilution series of a 441bp synthetic DNA fragment (IDT, USA) containing the Leishmania rDNA and hTNF target sequences and did not include DNA extraction. The DNA extraction and qPCR were repeated once when the hTNF Ct value was >32 or negative. Detection of amplified Leishmania rDNA resulted in defining the sample positive, except for Ct values  $\geq$ 40, which were classified as negative. Results were considered invalid if *Leishmania* rDNA qPCR was negative and the hTNF probe did not amplify. The technicians that performed the qPCR in Quito were unaware of the clinical characteristics of participants and the microscopy results.

#### Data analysis

Descriptive statistics (proportions, means, and medians, as appropriate) were used to describe the study population. We drew a directed acyclic graph (DAG) to illustrate the relation between the diagnostic tests that were used (i.e., smear slide microscopy and qPCR), the quantity of the pathogen itself, i.e., the amastigotes, or their genetic material (DNA) in the samples, and the target condition, being CL (S2 Fig). Based on this, we reasoned that the available data could be used to construct a two-class latent model with the two classes being "presence of CL" and "absence of CL". We used Bayesian latent class models, adjusted for conditional dependence, to estimate the sensitivity and specificity of qPCR and microscopy for the diagnosis of CL. We also estimated the likelihood ratios, the prevalence of CL in the study population correcting for their diagnostic accuracy and comparing to the observed prevalence, and the predictive values (negative and positive) with their 95% credible intervals (95%CrI). We used an informative prior for the specificity of microscopy (Beta (99,1) distribution with median of 99% (95%CrI 96%, 100%)) based on the known high specificity of the microscopy results. We also used an informative prior for qPCR specificity (Beta (97,3) with median of 97% (95%CrI 93%, 99%)) based on available evidence [13]. We present the sensitivity and specificity for both tests separately by region, the Amazon, and the Pacific (Model 1), after noticing the significant difference in sensitivity values between the two regions, not reflected in the pooled estimates (S1 Table). Sequentially, and as part of our secondary aim, we assessed the impact on diagnostic accuracy of pre-defined covariates or thus within pre-specified subgroups, i.e., defined by the socio-demographic and clinical variables. More specifically we investigated if the covariate health-seeking delay (cut-off 4 weeks) was an effect measure modifier versus confounder for the accuracy between the regions. We, therefore, allowed the model to estimate the sensitivity by covariate level (Model 2), while allowing for conditional dependence among CL positive subjects, in the subpopulations and the pooled sample test specificities. We further assessed the impact on diagnostic performance within pre-specified subgroups, i.e., defined by age (cut-off 20 years of age), gender, altitude of infection (cut-off 500m), body location of lesion (head and neck versus other location), assessing the predictive values in the subgroups. It was hypothesized that sensitivity for a single test would be clinically insufficient (threshold set at 80%) and that the NPV, which is dependent on the prevalence, would differ by 20 percentage points between the two regions. Only participants with paired samples, i.e., both a valid result for qPCR and microscopy, were included in the diagnostic accuracy analysis. For all statistics, medians and 95% CrIs were reported. Statistical significance was determined by a CrI of differences not including 0. To conclude and as a comparison, the accuracy of qPCR (and its 95% confidence interval (CI)) was estimated using microscopy as the reference standard from a two-by-two contingency table by region. We used R version 4.0 (R Foundation for Statistical Computing, Vienna, Austria, 2020), more specifically for carrying out Bayesian inference on the latent class model we used the rjags package.

#### Results

#### Participants

Participating centers enrolled 324 (50%) out of 646 eligible patients in the study (Fig 1). Four initially included participants were excluded because they had no cutaneous lesions. This resulted in a total of 320 included participants, a total of 188 (59%) participants from the Pacific and 132 (41%) participants from the Amazon region (Table 1). The mean age of the included participants was 26.8 years (range 0.1–88 years) and was higher in the Amazon (31.2 years) compared to the Pacific (23.7 years) region. A total of 100/188 (53%) participants included from the Pacific region were male compared to 85/132 (64%) from the Amazon. No Amerindian participants were included from the Pacific region compared to 87 (66%) from the Amazon. Median health-seeking delay in the Amazon was one month longer than in the Pacific (median 1 month). The lesion type (ulcer, nodule, or other) was known in 319/320 included participants who presented with a total of 322 lesion types. Three subjects presented with both ulcerated and nodular lesions and 16 with only nodular lesions. 290 participants presented with only ulcerated lesions were most frequently on the upper limbs (36%) and lower limbs





https://doi.org/10.1371/journal.pntd.0011745.g001

(30%). Amazon participants had fewer lesions on the head and neck (13%) compared to Pacific participants (25%). None of the participants had initiated anti-leishmanial treatment prior to the sampling.

#### Laboratory results

All filter papers arrived at the research laboratory in Quito and qPCR was done for every participant. Samples of five participants had an invalid qPCR result. For one participant, microscopy slides got lost in the including laboratory and could not be recovered. The including centers reported no adverse events during or after the sampling of the specimens.

After exclusion of the above mentioned six participants with an invalid or missing qPCR or microscopy test result, 314 participants are included in the diagnostic accuracy analysis. A total of 186/314 (59%) qPCR samples tested positive (49% in the Amazon and 64% in the Pacific) compared to 176/314 (56%) microscopy positive samples. Discordant results, microscopy positive while qPCR negative, were present in 14% and 17% of the cases in the Amazon and the Pacific; microscopy negative while qPCR positive results were present in 26% and 15% in the same regions. Ct values by region and microscopy results are presented in S2 Table. The qPCR detected  $1x10^{-9}$  ng/µL of synthetic DNA, which is equivalent to 4 copies of DNA in each 15uL reaction (S1 File). In 135/186 (73%) of the qPCR positive samples (71% Amazon, 74% Pacific) the causative *Leishmania* species could be determined, or in 46/129 (36%) and 89/185 (48%) of the total number of paired samples investigated in the Amazon and the Pacific, respectively [23]. Table 2 provides information on the species distribution by region, the sample internal control Ct values, and 18SrDNA Ct values by species. Microscopy positive varied by species, with 84% of *L. guyanensis* positive and only 42% of *L. braziliensis* positive.

Participant characteristic (N missing for variables)	Pacific	Amazon	Total <sup>c</sup>
Number of Cases (%)	188 (59)	132 (41)	320 (100)
General characteristics (1)			
Mean age in years (SD)	23.7 (18.8)	31.2 (20.1)	26.8 (19.7)
Male (%)	100 (53)	85 (64)	185 (58)
Mestizo (%)	188 (100)	45 (34)	233 (73)
Amerindian (%)	0 (0)	87 (66)	87 (27)
Clinical presentation (3)			
Median health-seeking delay in months (IQR)	1.0 (0.5-2.0)	2.0 (0.9-4.0)	1.0 (0.5-2.0)
Total number of lesion types presented (%) <sup>a</sup>	189 (59)	133 (41)	322 (100)
Ulcer (%)	170 (90)	123 (92)	293 (91)
Nodule (%)	9 (5)	10 (8)	19 (6)
Other (%)	10 (5)	0 (0)	10 (3)
Median number of lesions (IQR)	1 (1-2)	1 (1-1)	1 (1-2)
Median altitude of suspected place of infection in meters above sea level (IQR)	532 (219–779)	396 (278–647)	455 (274–739)
Body location of the lesion <sup>b</sup>			
Total number of body locations with lesions (%)	204 (59)	144 (41)	348 (100)
Head and neck (%)	51 (25)	18 (13)	69 (20)
Trunk (%)	22 (11)	25 (17)	47 (14)
Upper limbs (%)	77 (38)	49 (34)	126 (36)
Lower limbs (%)	54 (26)	52 (36)	106 (30)

 Table 1. Characteristics and clinical presentation of 320 suspected cutaneous leishmaniasis study participants from the Ecuadorian Pacific and Amazon regions by region and in total. Total number (%).

N = Number, SD = Standard Deviation, IQR = Inter Quartile Range

<sup>a</sup>The denominator is lesion types. Lesion type was unknown in one participant. Participants with several lesion types were counted more than once.

<sup>b</sup>The denominator is body locations with lesions, participants with lesions on different body regions were counted more than once

<sup>c</sup>General description of 6 participants with missing paired samples is included in the table. 5 participants had an invalid PCR result, 1 had a lost microscopy sample. A total of 185 participants in the Amazon and 129 in the Pacific provided paired samples for the accuracy analysis.

https://doi.org/10.1371/journal.pntd.0011745.t001

#### Diagnostic accuracy estimates of qPCR on DNA extracted from filter paper and microscopy

Applying our model (Model 1) to the population of the Amazon region, we found a sensitivity of qPCR of 68% (95%CrI 49;82) with a specificity of 97% (95%CrI 93;99), and a sensitivity of 51% (95%CrI 36;66) with a specificity of 99% (95%CrI 96;100) for microscopy. In the Pacific region, qPCR sensitivity and specificity were 73% (95%CrI 63;83) and 97% (95%CrI 93;99), respectively. Microscopy sensitivity was 76% (95%CrI 65;86) with a specificity of 99% (95%CrI 96;100) (Table 3). Microscopy sensitivity was statistically significantly lower in the Amazon compared to the Pacific (-24.9 percentage points (95%CrI -43.5; -6.7). Differences between qPCR and microscopy sensitivity were also largest in the Amazon (-16 percentage points (95% CrI -31;-2) and statistically significant. Estimating the diagnostic accuracy of qPCR and microscopy by health-seeking delay (Model 2) separately for the two regions, there is no evidence of an effect in the Amazon on qPCR or microscopy and a non-statistically significant effect on the microscopy in the Pacific region, with 8.1 percentage points (95%Cr -26.9;10.2) lower sensitivity in cases presenting more than 4 weeks after reported lesion appearance

	L. guyanensis (N = 102) <sup>a</sup>	L. braziliensis (N = 26) <sup>a</sup>	L. lainsoni $(N = 7)^a$	All samples (N = 135)
Microscopy positive (%)				
Total	86 (84)	11 (42)	6 (86)	103 (76)
Amazon	13 (68)	7 (33)	5 (83)	25 (54)
Pacific <sup>b</sup>	73 (88)	4 (80)	1 (100)	78 (88)
Median hTNF Ct (IQR) <sup>c</sup>	29.4 (27.3-31.0)	27.1 (25.7–28.9)	30.6 (29.9)	29.0 (26.8-30.8)
Median <i>Leishmania</i> 18rDNA Ct (IQR) <sup>c</sup>	29.7 (27.7–32.5)	31.7 (29.8-35.2)	34.4 (27.7–35.3)	30.3 (27.8-33.3)

Table 2. Proportion microscopy positives by region and cycle threshold values of internal controls and of Leishmania 18rDNA in the 135 qPCR positive cases with *Leishmania* species identification, by species and in all samples combined.

hTNF: human Tumor Necrosis Factor, Ct: Cycle Threshold, IQR: Interquartile Range

<sup>a</sup> Leishmania species was determined in Leishmania 18SrDNA positive samples by sequencing a gene fragment that codes for the Leishmania Cytochrome B enzyme. <sup>b</sup>By comparison, 81% (N = 83), 19% (N = 5) and 14% (N = 1) of the as *L. guyanensis*, *L. braziliensis*, *L. lainsoni* identified species in the total of investigated samples are from the Pacific region. A total of 66% of the identified species is from the Pacific region, with the remaining being from the Amazon.

<sup>c</sup> hTNF was applied as internal control for sample taking and DNA extraction in a duplex qPCR together with *Leishmania* rDNA. Ct values have a logarithmic relationship with DNA concentrations and lower Ct values indicate higher DNA copy numbers.

https://doi.org/10.1371/journal.pntd.0011745.t002

(Table 4). Using microscopy as the perfect reference standard, we estimated a sensitivity of 48% (95% confidence Interval (CI) 35;61) and specificity of 72% (95% CI 59;82) for qPCR in the Amazon and a sensitivity of 78% (95%CI 69;85) and specificity of 52% (95% CI 39;64) in the Pacific (Table 3).

	Model 1 <sup>a</sup> in subpopulation of the Amazon	Model 1 <sup>a</sup> in subpopulation of the Pacific	two-by-two-table in subpopulation of the Amazon <sup>b</sup>	two-by-two-table in subpopulation of the Pacific <sup>c</sup>
Sensitivity qPCR	68.0% (49.1;82.4)	73.4% (62.7;82.7)	63.3% (48.3;76.6)	75.2% (66.7;82.5)
Sensitivity Microscopy	51.2% (35.9;65.5)	76.4% (65.0;85.6)	-	-
Difference sensitivity <sup>d</sup>	-16.0 (-31.4;-1.7)	2.9 (-6.5;12.2)	-	-
Specificity qPCR	97.2% (92.6;99.4)	97.2% (92.8;99.4)	57.5% (45.9;68.5)	55.0% (41.6;67.9)
Specificity Microscopy	99.3% (96.1;100)	99.3% (96.1;100)	-	-
Difference specificity <sup>d</sup>	1.8 (-1.7;6.6)	1.8 (-1.7;6.4)	-	-
LR+ qPCR	24 (9;107)	27 (10;113)	1.5 (1.1;2.2)	1.7 (1.2;2.3)
LR+ Microscopy	71 (13;1852)	105 (19;2400)	-	-
LR- qPCR	0.3 (0.2;0.5)	0.3 (0.2;0.4)	0.6 (0.4;1)	0.5 (0.3;0.7)
LR- Microscopy	0.5 (0.3;0.6)	0.2 (0.1;0.4)	-	-

Table 3. Diagnostic accuracy estimates with their 95% credible interval for qPCR and microscopy using latent class analysis and two-by-two table calculation (95% confidence interval).

LR+: Positive Likelihood Ratio: true-positive proportion/false-positive proportion; LR-: false-negative proportion/true-negative proportion. A test with a LR+ of >10 is considered useful to rule in a diagnosis when a test is positive, while a test with a LR- <0.1 is considered useful to exclude a diagnosis when a test is negative. <sup>a</sup>Latent class analysis with two latent classes, using the data of one joint population and informative prior for specificity microscopy with a beta distribution (99,1) and informative prior for specificity qPCR with a beta distribution (97,3). The model allows for conditional dependency between the two tests' sensitivities and specificities. <sup>b</sup>Two-by-two contingency table of samples from the Amazon region, using microscopy as the reference standard: true positives N = 31; false positives N = 34, false negatives N = 18, and true negatives N = 46.

 $^{\circ}$ Two-by-two contingency table of samples from the Pacific region, using microscopy as the reference standard: true positives N = 94; false positives N = 27, false negatives N = 31 and true negatives N = 33.

<sup>d</sup>Difference between the estimate for qPCR and microscopy in percentage points

https://doi.org/10.1371/journal.pntd.0011745.t003

	Model 1 <sup>a</sup> in subpopulation of	Model 1 <sup>a</sup> in subpopulation of	two-by-two-table in subpopulation	two-by-two-table in
	the Amazon	the Pacific	of the Amazon <sup>o</sup>	subpopulation of the Pacific
Sensitivity qPCR	68.0% (49.1;82.4)	73.4% (62.7;82.7)	63.3% (48.3;76.6)	75.2% (66.7;82.5)
Sensitivity Microscopy	51.2% (35.9;65.5)	76.4% (65.0;85.6)	-	-
Difference sensitivity <sup>d</sup>	-16.0 (-31.4;-1.7)	2.9 (-6.5;12.2)	-	-
Specificity qPCR	97.2% (92.6;99.4)	97.2% (92.8;99.4)	57.5% (45.9;68.5)	55.0% (41.6;67.9)
Specificity Microscopy	99.3% (96.1;100)	99.3% (96.1;100)	-	-
Difference specificity <sup>d</sup>	1.8 (-1.7;6.6)	1.8 (-1.7;6.4)	-	-
LR+ qPCR	24 (9;107)	27 (10;113)	1.5 (1.1;2.2)	1.7 (1.2;2.3)
LR+ Microscopy	71 (13;1852)	105 (19;2400)	-	-
LR- qPCR	0.3 (0.2;0.5)	0.3 (0.2;0.4)	0.6 (0.4;1)	0.5 (0.3;0.7)
LR- Microscopy	0.5 (0.3;0.6)	0.2 (0.1;0.4)	-	-
	Model 1 <sup>a</sup> in subpopulation of the Amazon	Model 1 <sup>a</sup> in subpopulation of the Pacific	two-by-two-table in subpopulation of the Amazon <sup>b</sup>	two-by-two-table in subpopulation of the Pacific <sup>c</sup>
Sensitivity qPCR	68.0% (49.1;82.4)	73.4% (62.7;82.7)	63.3% (48.3;76.6)	75.2% (66.7;82.5)
Sensitivity Microscopy	51.2% (35.9;65.5)	76.4% (65.0;85.6)	-	-
Difference sensitivity <sup>d</sup>	-16.0 (-31.4;-1.7)	2.9 (-6.5;12.2)	-	-
Specificity qPCR	97.2% (92.6;99.4)	97.2% (92.8;99.4)	57.5% (45.9;68.5)	55.0% (41.6;67.9)
Specificity Microscopy	99.3% (96.1;100)	99.3% (96.1;100)	-	-
Difference specificity <sup>d</sup>	1.8 (-1.7;6.6)	1.8 (-1.7;6.4)	-	-
LR+ qPCR	24 (9;107)	27 (10;113)	1.5 (1.1;2.2)	1.7 (1.2;2.3)
LR+ Microscopy	71 (13;1852)	105 (19;2400)	-	-
LR- qPCR	0.3 (0.2;0.5)	0.3 (0.2;0.4)	0.6 (0.4;1)	0.5 (0.3;0.7)
LR- Microscopy	0.5 (0.3;0.6)	0.2 (0.1;0.4)	-	-

Table 4. Diagnostic accuracy estimates by delay in presentation for qPCR and microscopy using latent class analysis Model 2.

LR+: Positive Likelihood Ratio: true-positive proportion/false-positive proportion; LR-: false-negative proportion/true-negative proportion. A test with a LR+ of >10 is considered useful to rule in a diagnosis when a test is positive, while a test with a LR- <0.1 is considered useful to exclude a diagnosis when a test is negative. <sup>a</sup>Latent class analysis with two latent classes, using the data of two subpopulations, i.e., the binary variable delay in presentation, allowing for differing sensitivities by stratum, and informative prior for specificity microscopy with a beta distribution (99,1) and informative prior for specificity qPCR with a beta distribution (97,3). The model allows for conditional dependency between the two tests sensitivities and between the subpopulation sensitivities, and two tests specificities. <sup>b</sup>Difference between the estimate for qPCR and microscopy in percentage points

https://doi.org/10.1371/journal.pntd.0011745.t004

# Prevalence estimation, PPV, NPV, and effect of demographic and clinical characteristics on diagnostic performance

The prevalence of CL in the sampled cases was 73% (95%CrI 58;97) in the Amazon region and 88% (95%CrI 78;99) in the Pacific region using Model 1, correcting for test accuracy. In the Amazon, the addition of qPCR test positive cases to those identified by microscopy (observed prevalence microscopy positives: 38.0%) can increase the observed prevalence with 26.4 (95%CrI 19.3;34.4) percentage points. In the Pacific, the prevalence increases 14.6 (95%CrI 10.1;20.2) percentage points, also improving the diagnostic yield (Table 5). PPV and NPV are presented by region and by the different covariates age, altitude of infection, and lesion body location in with their estimates in the S3 Table, together with the estimates of covariate stratified sensitivities and specificities by region. NPV in the Amazon region was overall 54% (95%CrI 5;77) and 44% (95%CrI 4;65) for qPCR and microscopy respectively while being 34% (95%CrI 3;58) and 37 (95%CrI 3;63) in the Pacific region. Table 5. Disease prevalence estimates in the sampled population correcting for the diagnostic accuracy of qPCR and microscopy (true prevalence); prevalence observed using qPCR and microscopy separately and the potential additional prevalence detected by adding qPCR positive cases to microscopy negative cases, by region.

	Amazon	Pacific
Estimated true prevalence (Model 1) <sup>a</sup>	72.6% (57.9; 97.3)	87.7% (78.1;99.1)
Observed prevalence of microscopy positive	38.0% (30.0;46.6)	67.6% (60.6;74.0)
Observed prevalence of qPCR positive	50.4% (41.8;58.9)	65.4% (58.4;72.0)
Observed prevalence of either microscopy or qPCR positive <sup>b</sup>	64.3% (55.8;72.2)	82.3% (76.2;87.2)
Additional prevalence (in percentage points) diagnosed by adding qPCR to microscopy prevalence (microscopy negatives, qPCR positives)	26.4 (19.3;34.4)	14.6 (10.1;20.2)

<sup>a</sup>The prevalence is estimated taking both test results and their imperfect accuracy into account and provides the estimated prevalence of CL disease, the target condition under investigation in the study sample.

<sup>b</sup>The sum is lower than the true prevalence estimates, given this is a prevalence based on observed positives and not corrected for imperfect accuracy. In this scenario, both false positive microscopy cases and false positive qPCR cases contribute to the prevalence and cases with positive qPCR and microscopy agreement are not double counted.

https://doi.org/10.1371/journal.pntd.0011745.t005

#### Discussion

In this diagnostic accuracy study, using latent class analysis in participants presenting with suspected CL diagnosis in two regions in Ecuador, we found that the sensitivity of qPCR on DNA extracted from filter paper and smear slide microscopy is considerably lower in the Amazon region than in the Pacific region. This difference could not be explained by differences in health-seeking delay. In addition, different participants were diagnosed with either test. The specificity point estimates were 97% for qPCR and 99% for microscopy, using informative priors. The NPV reached its highest value in the lower prevalence region of the Amazon (prevalence 73%) both for qPCR and microscopy with a point estimate of 54% and 44% respectively, still lower than needed to confidentially consider the negative test as the proof of absence of CL in the participant. Adding the qPCR test and including qPCR positive cases as confirmed cases of CL would however increase the detection of cases with 15 to 26 percentage points, depending on the region. Other socio-demographic or clinical characteristics did not provide evidence to change the posterior probability of disease, more particularly, disease absence.

This study has several limitations. First, recruitment and thus the sample size suffered from the study being performed during the COVID-19 pandemic. During the COVID-19 pandemic, treatment supply at national level was interrupted temporarily and therefore the care for suspected CL patients was postponed in an unknown percentage. Additionally, individuals might have been afraid or had competing interests leading to lower participation rates than anticipated. We do not assume that participation was differential to qPCR or microscopy results, however. Because of the heterogeneous effect by region, the stratified sample sizes are relatively small, resulting in wider CrI. Secondly, with only two tests performed, we cannot further refine our model or assess the underlying mechanisms of the disagreement between the microscopy tests and qPCR. Third, the study population assessed might not be transportable to other environments, most specifically concerning their distribution of covariates. Additionally, the distribution of the covariates in the study population is conditional on the testing strategies that are in use in the different zones in Ecuador, which can lead to selection bias. Prevalence distributions can shift when the pre-test probabilities change over time or by center, alike the probability that an individual presenting with a lesion has an alternative diagnosis compared to CL. Fourth, the microscopic observations were made by different lab technicians leading possibly to a bias in the results, with most likely introduction of more false negatives, thereby decreasing the sensitivity of microscopy.

In this study, we aimed to answer a diagnostic question relevant to the correct diagnosis of CL when no good reference standard is available and where there is a lack of functional diagnostic tools for its diagnosis. The use of the LCA allowed the estimation of the sensitivity and specificity for qPCR and microscopy in the same data set, therefore estimating the accuracy of the test in use and the potential alternative or add-on test. By using an LCA we did not assume that microscopy has perfect sensitivity or specificity. We used priors for the specificity of both tests. While single microscopy tests can suffer from artifacts being recognized as amastigotes, the results, as in the current practice, are only positive when the presence of amastigotes is confirmed by the central laboratory, through second reading.

When compared to several other studies assessing the accuracy of PCR on DNA extracted from filter paper, our study found a lower sensitivity and lower positive agreement with microscopy [28,31]. On the other hand, our findings are consistent with a study of >700 Palestinian CL suspects that found a limited sensitivity of PCR on DNA extracted from filter paper and disagreement with smear slide microscopy [32]. Such a disagreement may be caused by a number of issues: First, because amastigotes are distributed unevenly across skin layers, the sample Leishmania DNA concentration may be affected by the sampling technique [33]. In our study, the sample for microscopy was obtained by scraping and the sample for qPCR by imprinting the lesion on filter paper. In other studies, lesion imprints on filter paper followed by PCR have shown better accuracy in comparison to scrapings followed by microscopy, however, also failed to detect Leishmania DNA in 8-17% of the proven microscopy positive samples [34-36]. A direct comparison of lesion imprints and scrapings on filter paper could reveal whether the sampling technique resulted in false negatives. Second, the sampling site may cause heterogeneity, but in this study, both the filter paper imprints and scrapings were taken from the lesion's inner border [37]. Third, the handling and transport of the filter papers in remote tropical forest areas under uncontrolled conditions may still have affected the DNA quality. Fourth, when compared to other methods, the in our study used Chelex-based DNA extraction can result in more DNA extracted [38]. This may help detect *Leishmania* DNA, but abundant non-Leishmania DNA and contamination may also interfere with amplification [31,38]. Chelex has been applied in a limited number of studies to extract DNA from filter paper for the detection of CL, so an optimization study is recommended. Fifth, the Leishmania 18S qPCR used in this study has been validated in South American CL samples and is expected to detect the Ecuadorian endemic species [39]. Nonetheless, because the qPCR lacks a reverse transcriptase step, it does not amplify RNA, which may be more abundant in the samples than DNA. We recommend that in the future, the qPCR includes a reverse transcriptase step, as described by van der Meide et al. [29]. Finally, grading of the parasite density of Leishmania positive microscopy slides might have clarified an association between parasitemia and qPCR false negativity in our study and is recommended for future studies. We found heterogeneity in the sensitivity of microscopy by region in Ecuador. The country's mainland is divided from north to south by the Andean highlands where CL is rare. CL clusters occur in the northwestern (Pacific) and the entire eastern (Amazon) region [4]. The probable reservoir, vector, and infecting Leishmania species differ by these regions. L. guyanensis is the prevalent species in participants from the Pacific region and a mix of L. guyanensis, L. braziliensis, L. lainsoni, and L. naiffi species is prevalent in the Amazon [4,12,40]. Detailed surveillance data by region are however not available. Additionally, participant presentation (age, health-seeking delay, and body location of lesions) and quality of life of CL-suspected patients are region dependent [5,12]. Participants from the Pacific region were included in the three cantons with Ecuador's highest burden of CL (212-464 cases per 10.000 inhabitants). The Amazon participants were included from cantons with lower CL burden (17-212 cases per 10.000 inhabitants) [4]. The differences in participant presentation (lower age, shorter health-seeking delay, and more

lesions on the head and neck in the Pacific region) and estimated CL prevalence as found in this study are in agreement with former publications [41,42]. However, this is the first publication to reflect on the diagnostic accuracy of both tests by region. The diagnostic program for leishmaniasis is Ecuador wide with uniform training procedures for microscopy technicians, as was the training of technicians involved in the filter paper sampling for this study [6]. Nevertheless, the lower prevalence of leishmaniasis in the Amazon may have resulted in fewer samples per technician, less experience, and, as a result, a lower sensitivity of smear slide microscopy [4]. Additional information about the lesions (diameter, wetness, infected or not) was not collected in this study, which could have helped to explain differences [43]. The causative Leishmania species, a prolonged period of health-seeking delay, and/or participant variables such as lesion location and age could all have influenced test performance [44]. This study's strength is that it relied on existing diagnostic structures and thus reflects actual clinical practice. This has the limitation of not allowing us to draw conclusions about the determinants of test accuracies which should be addressed in future research. Because CL is more prevalent in the Pacific than in the Amazon, professionals in the Pacific (more experienced) may have sent fewer non-CL participants for Leishmania testing, resulting in a higher true prevalence. The differences in NPV depend on both test accuracy and estimated true prevalence and they remained below 80% in both the Amazon and Pacific, making it impossible to exclude CL after receiving a negative test from a suspected case.

The use of socio-demographic and clinical characteristics and their integration in the diagnostic pathway has been suggested, given this information is easily available also in very remote regions, is known on presentation, and does not require costly funding. Assessing the sensitivity and specificity of these characteristics, however, does not inform us how to interpret a positive or negative result of a qPCR or microscopy test and make a more informed treatment decision. Estimating the predictive values and likelihood ratios does allow their use in clinical decision-making. We showed that the sociodemographic and clinical characteristics as assessed do not improve the posterior probability to exclude the diagnosis of CL, i.e., the updated probability to not have the disease when the test is negative and this additional information is taken into account [45]. While, for example, exposure to the vector, i.e., in Ecuador the sandfly of the subfamily Phlebotominae genus *Lutzomyia*, has historically been defined by sex and/or gender, we did not see a difference in the prevalence of CL cases in males versus females who were tested for CL. As an additional remark, the covariates were dichotomized, which will result in information loss and might obscure real differences if we had chosen different cut-offs or had used the continuous variable.

Lessons learned from the results of this study for clinical practice and use of the diagnostic tests qPCR and microscopy are the following: First, a larger study is necessary to be able to decrease the uncertainty around the estimates. Second, data on the first microscopy and agreement with the confirmative reading will provide additional information. Our analysis can only be part of the general approaches to assess the true accuracy of tests used in the clinical diagnosis of CL. The addition and inclusion of direct PCR sampling and different sampling procedures can however optimize the specimen for PCR and its diagnostic yield, as is also the recommended diagnostic strategy as published in the IDSA guidelines [46]. Using microscopy solely for diagnosis maintains the status quo of under ascertainment and thus under treatment of participants with CL.

#### Conclusions

The accuracy of the diagnostic tests evaluated for the diagnosis of CL seems to be unsatisfactory. Smear slide microscopy suffers from being insufficiently sensitive and being region dependent while having good specificity. None of the investigated tests, qPCR nor microscopy alone, has a sufficient performance to confidently rule out CL in the presenting participants. We recommend further studies on qPCR and microscopy accuracy to avoid increased morbidity and a sustained burden of DALY's related to CL, due to participants who remained without a diagnosis and therefore untreated. An additional diagnostic test, either microscopy or qPCR, seems necessary to improve the overall sensitivity of the diagnostic strategy.

#### Supporting information

**S1 STARD-BLCM Checklist. Completed STARD-BLCM Checklist.** (DOCX)

**S1 Fig. Altitude map of Ecuador with geographic regions and participating health centers** (**N** = 16). Black dots indicate major cities. Red dots indicate participating health center locations in the Pacific region: 1: Puerto Quito, 2: Pedro Vicente Maldonado, 3: San Miguel de Los Bancos, and in the Amazon region: 4: Tena Hospital, 5: Paushiyaku, 6: Satelital Tena, 7: Puerto Napo, 8: Misahualli, 9: Chontapunta, 10: Arosemena Tola, 11: Shell hospital, 12: Puyo hospital, 13: Tuutinentza, 14: Ipiak, 15: Wasakentsa, 16: Wachirpas. Copyright: The image is adapted from Wikipedia by the authors and is available under the Creative Commons CC0 1.0 Universal Public Domain Dedication [47].

(TIF)

S2 Fig. Directed acyclic graph (DAG)—Graphical representation of the study question.

The target condition under study is cutaneous leishmaniasis. Amastigotes or their genetic material (DNA) in the wound under suspicion are the measurands. The tests under evaluation are qPCR on DNA extracted from filter paper and direct microscopy after staining of a sample taken by wound scraping. The figure represents a DAG of the study question. We define two latent classes in this accuracy question, being (i) cutaneous leishmaniasis disease positive and (ii) disease status negative. Covariates potentially associated with a difference in prevalence of disease were assessed. Geographical region (Amazon versus Pacific region) was found an important factor. Other covariates were investigated separately by geographical region. Altitude of infection (500 m cut-off), body location of the lesion (head or neck versus elsewhere), health-seeking delay (cut-off 4 weeks) and age (younger or older than 20 years old) were only minimally different. Disease was equally distributed among both sexes. (TIF)

**S1 Table. Diagnostic accuracy estimates for qPCR and microscopy with 95% credible Intervals using latent class analysis using region pooled data.** LR: likelihood ratio; LR+: positive likelihood ratio; LR-: negative likelihood ratio. <sup>a</sup>Latent class analysis with two latent classes, using the data of one joint population and informative prior for specificity microscopy with a beta distribution (99,1) and for specificity qPCR with a beta distribution (97,3). <sup>b</sup>Difference between estimate for qPCR and for microscopy in percentage points (DOCX)

**S2 Table. Characteristics of filter paper samples of 314 further analyzed cases.** qPCR: quantitative Polymerase Chain Reaction, hTNF: human Tumor Necrosis Factor, Ct: Cycle Threshold, IQR: Interquartile Range. <sup>a</sup> hTNF was applied as internal control for sample taking and DNA extraction in a duplex qPCR together with *Leishmania* rDNA. Ct values have a logarithmic relationship with DNA concentrations and lower Cts indicate higher DNA copy numbers. <sup>b</sup> *Leishmania* species was determined in *Leishmania* 18SrDNA qPCR positive samples by sequencing a gen fragment that codes for the *Leishmania* Cytochrome B enzyme. (DOCX)

S3 Table. Prevalence, PPV and NPV with 95% credible Interval, by covariate using LCA<sup>a</sup>.

<sup>a</sup>Models use a beta distribution for the priors for sensitivity and specificity, including the informative prior for microscopy specificity of 99% and qPCR specificity of 97%. (DOCX)

**S1** File. Limit of detection and standard curve of the qPCR reaction. (DOCX)

**S2** File. Study data of 320 participants. (XLSX)

#### Acknowledgments

We would like to thank all of the study participants and staff for their contributions and participation.

#### **Author Contributions**

**Conceptualization:** Jacob M. Bezemer, Joanna Merckx, Manuel Calvopiña, Henry J. C. de Vries, Henk D. F. H. Schallig, Mariska M. G. Leeflang, Nandini Dendukuri.

Data curation: Jacob M. Bezemer.

Formal analysis: Joanna Merckx, Nandini Dendukuri.

Funding acquisition: Jacob M. Bezemer.

Investigation: Jacob M. Bezemer, Byron P. Freire Paspuel, Manuel Calvopiña.

Methodology: Jacob M. Bezemer, Joanna Merckx.

Project administration: Jacob M. Bezemer, Joanna Merckx.

**Resources:** Jacob M. Bezemer, Joanna Merckx, Byron P. Freire Paspuel, Manuel Calvopiña, Henry J. C. de Vries, Henk D. F. H. Schallig, Mariska M. G. Leeflang, Nandini Dendukuri.

- Supervision: Manuel Calvopiña, Henry J. C. de Vries, Henk D. F. H. Schallig, Mariska M. G. Leeflang, Nandini Dendukuri.
- Validation: Byron P. Freire Paspuel, Nandini Dendukuri.

Visualization: Jacob M. Bezemer, Joanna Merckx.

- Writing original draft: Jacob M. Bezemer, Joanna Merckx.
- Writing review & editing: Byron P. Freire Paspuel, Manuel Calvopiña, Henry J. C. de Vries, Henk D. F. H. Schallig, Mariska M. G. Leeflang, Nandini Dendukuri.

#### References

- Kumari D, Perveen S, Sharma R, Singh K. Advancement in leishmaniasis diagnosis and therapeutics: An update. European Journal of Pharmacology. 2021; 910:174436. https://doi.org/10.1016/j.ejphar. 2021.174436 PMID: 34428435
- de Vries HJ, Schallig HD. Cutaneous Leishmaniasis: A 2022 Updated Narrative Review into Diagnosis and Management Developments. American Journal of Clinical Dermatology. 2022:1–18. https://doi.org/ 10.1007/s40257-022-00726-8 PMID: 36103050
- Saeedi M. Leishmaniasis Geneva: WHO; 2022. <u>https://www.who.int/news-room/fact-sheets/detail/leishmaniasis</u>. Accessed: 25 January 2022.
- 4. Henríquez-Trujillo AR, Coral-Almeida M, Hinojosa MC. The burden of cutaneous and mucocutaneous leishmaniasis in Ecuador (2014–2018), a national registry-based study. bioRxiv. 2019:751446.

- Bezemer JM, Calvopiña Hinojosa M, Corral Zabala AE, Ortega Pérez F, Vargas Román VC, Schallig HDFH, de Vries HJC. Quality of life of cutaneous leishmaniasis suspected patients in the Ecuadorian Pacific and Amazon regions: a cross sectional study. BMC infectious diseases. 2022; 22(1):1–9.
- Ministerio De Salud Pública. Manual de procedimientos del Subsistema alerta acción SIVE–ALERTA. Second ed. Quito: MSP; 2013.
- Aviles H, Belli A, Armijos R, Monroy FP, Harris E. PCR detection and identification of Leishmania parasites in clinical specimens in Ecuador: a comparison with classical diagnostic methods. The Journal of parasitology. 1999:181–7. PMID: 10219292
- Calvopina M, Guevara A, Gomez E, Guderian R, Mimori T, Hashiguchi Y. A comparative study of conventional and PCR-based diagnosis of cutaneous leishmaniasis in Ecuador. Research Report Series; 1997.
- Matsumoto T, Hashiguchi Y, Gomez E, Calvopina M, Nonaka S, Saya H, Mimori T. Comparison of PCR results using scrape/exudate, syringe-sucked fluid and biopsy samples for diagnosis of cutaneous leishmaniasis in Ecuador. Transactions of the Royal Society of Tropical Medicine and Hygiene. 1999; 93 (6):606–7. https://doi.org/10.1016/s0035-9203(99)90065-2 PMID: 10717745
- Chico ME, Guderian RH, Cooper PJ, Armijos R, Grogl M. Evaluation of a direct immunofluorescent antibody (DIFMA) test using Leishmania genus-specific monoclonal antibody in the routine diagnosis of cutaneous leishmaniasis. Revista da Sociedade Brasileira de Medicina Tropical. 1995; 28:99–103. https://doi.org/10.1590/s0037-86821995000200002 PMID: 7716331
- Mimori T, Matsumoto T, Calvopiña M, Gomez E, Saya H, Katakura K, et al. Usefulness of sampling with cotton swab for PCR-diagnosis of cutaneous leishmaniasis in the New World. Acta tropica. 2002; 81 (3):197–202. https://doi.org/10.1016/s0001-706x(01)00215-7 PMID: 11835896
- Hashiguchi Y, Velez LN, Villegas NV, Mimori T, Gomez EAL, Kato H. Leishmaniases in Ecuador: Comprehensive review and current status. Acta tropica. 2017; 166:299–315. https://doi.org/10.1016/j. actatropica.2016.11.039 PMID: 27919688
- Pena HP, Belo VS, Xavier-Junior JCC, Teixeira-Neto RG, Melo SN, Pereira DA, et al. Accuracy of diagnostic tests for American tegumentary leishmaniasis: a systematic literature review with meta-analyses. Tropical medicine & international health: TM & IH. 2020; 25(10):1168–81. <u>https://doi.org/10.1111/tmi.</u> 13465 PMID: 32677284
- Smit PW, Elliott I, Peeling RW, Mabey D, Newton PN. An overview of the clinical use of filter paper in the diagnosis of tropical diseases. The American journal of tropical medicine and hygiene. 2014; 90 (2):195. https://doi.org/10.4269/ajtmh.13-0463 PMID: 24366501
- Bezemer JM, Meesters K, Naveda CL, Machado PR, Calvopiña M, Leeflang MM, et al. Clinical criteria for Mucosal Leishmaniasis diagnosis in rural South America: A systematic literature review. PLoS neglected tropical diseases. 2022; 16(8):e0010621. https://doi.org/10.1371/journal.pntd.0010621 PMID: 35947553
- Weigle KA, Escobar M, Arias AL, Martinez F, Rojas C. A clinical prediction rule for American cutaneous leishmaniasis in Colombia. International Journal of Epidemiology. 1993; 22(3):548–58. <u>https://doi.org/ 10.1093/ije/22.3.548</u> PMID: 8359974
- Altman DG, Bland JM. Statistics Notes: Diagnostic tests 2: predictive values. Bmj. 1994; 309 (6947):102.
- Cantanhêde LM, Mattos CB, Cruz AK, Ikenohuchi YJ, Fernandes FG, Medeiros E, et al. Overcoming the Negligence in Laboratory Diagnosis of Mucosal Leishmaniasis. Pathogens. 2021; 10(9). <u>https://doi.org/10.3390/pathogens10091116</u> PMID: 34578149
- Dendukuri N, Schiller I, De Groot J, Libman M, Moons K, Reitsma J, Van Smeden M. Concerns about composite reference standards in diagnostic research. bmj. 2018;360. <u>https://doi.org/10.1136/bmj.j5779</u> PMID: 29348126
- MacLean EL, Kohli M, Köppel L, Schiller I, Sharma SK, Pai M, et al. Bayesian latent class analysis produced diagnostic accuracy estimates that were more interpretable than composite reference standards for extrapulmonary tuberculosis tests. Diagnostic and prognostic research. 2022; 6(1):1–10.
- Boelaert M, Aoun K, Liinev J, Goetghebeur E, Van Der Stuyft P. The potential of latent class analysis in diagnostic test validation for canine Leishmania infantum infection. Epidemiology & Infection. 1999; 123 (3):499–506. https://doi.org/10.1017/s0950268899003040 PMID: 10694163
- 22. World Health Organization. Ending the neglect to attain the Sustainable Development Goals: a road map for neglected tropical diseases 2021–2030. 2020.
- Bezemer JM, Freire-Paspuel BP, Schallig HD, de Vries HJ, Calvopiña M. Leishmania species and clinical characteristics of Pacific and Amazon cutaneous leishmaniasis in Ecuador and determinants of health-seeking delay: a cross-sectional study. BMC infectious diseases. 2023; 23(1):395. <u>https://doi.org/10.1186/s12879-023-08377-8 PMID: 37308815</u>

- Kostoulas P, Nielsen SS, Branscum AJ, Johnson WO, Dendukuri N, Dhand NK, et al. STARD-BLCM: Standards for the Reporting of Diagnostic accuracy studies that use Bayesian Latent Class Models. Preventive veterinary medicine. 2017; 138:37–47. https://doi.org/10.1016/j.prevetmed.2017.01.006 PMID: 28237234
- 25. Conteron E. Prevalencia de leishmaniasis en el area II de Pastaza, asociados a factores de riesgo que influyen en el desarrollo de la enfermedad. Ambato: Universidad Tecnica de Ambato; 2015.
- 26. Bastidas CA, Villacrés-Granda I, Navarrete D, Monsalve M, Coral-Almeida M, Cifuentes SG. Antibiotic susceptibility profile and prevalence of mecA and lukS-PV/lukF-PV genes in Staphylococcus aureus isolated from nasal and pharyngeal sources of medical students in Ecuador. Infection and Drug Resistance. 2019; 12:2553. https://doi.org/10.2147/IDR.S219358 PMID: 31496767
- Panda BB, Meher AS, Hazra RK. Comparison between different methods of DNA isolation from dried blood spots for determination of malaria to determine specificity and cost effectiveness. Journal of Parasitic Diseases. 2019; 43(3):337–42. https://doi.org/10.1007/s12639-019-01136-0 PMID: 31406397
- Miranda A, Saldaña A, González K, Paz H, Santamaría G, Samudio F, Calzada J. Evaluation of PCR for cutaneous leishmaniasis diagnosis and species identification using filter paper samples in Panama, Central America. Transactions of the Royal Society of Tropical Medicine and Hygiene. 2012; 106 (9):544–8. https://doi.org/10.1016/j.trstmh.2012.05.005 PMID: 22818741
- 29. van der Meide W, Guerra J, Schoone G, Farenhorst M, Coelho L, Faber W, et al. Comparison between quantitative nucleic acid sequence-based amplification, real-time reverse transcriptase PCR, and real-time PCR for quantification of Leishmania parasites. Journal of Clinical Microbiology. 2008; 46(1):73–8. https://doi.org/10.1128/JCM.01416-07 PMID: 17959763
- Weirather JL, Jeronimo SM, Gautam S, Sundar S, Kang M, Kurtz MA, et al. Serial quantitative PCR assay for detection, species discrimination, and quantification of Leishmania spp. in human samples. Journal of clinical microbiology. 2011; 49(11):3892–904. https://doi.org/10.1128/JCM.r00764-11 PMID: 22042830
- Mota CA, Venazzi EAS, Zanzarini PD, Aristides SMA, Lonardoni MVC, Silveira TGV. Filter paper performance in PCR for cutaneous leishmaniasis diagnosis. Revista da Sociedade Brasileira de Medicina Tropical. 2020;54. https://doi.org/10.1590/0037-8682-0047-2020 PMID: 33338104
- Al-Jawabreh A, Dumaidi K, Ereqat S, Nasereddin A, Azmi K, Al-Jawabreh H, et al. A comparison of the efficiency of three sampling methods for use in the molecular and conventional diagnosis of cutaneous leishmaniasis. Acta tropica. 2018; 182:173–7. <u>https://doi.org/10.1016/j.actatropica.2018.03.001</u> PMID: 29522706
- 33. Sevilha-Santos L, Dos Santos Júnior ACM, Medeiros-Silva V, Bergmann JO, da Silva EF, Segato LF, et al. Accuracy of qPCR for quantifying Leishmania kDNA in different skin layers of patients with American tegumentary leishmaniasis. Clinical microbiology and infection: the official publication of the European Society of Clinical Microbiology and Infectious Diseases. 2019; 25(2):242–7. https://doi.org/10. 1016/j.cmi.2018.04.025 PMID: 29730222
- Boggild AK, Valencia BM, Espinosa D, Veland N, Ramos AP, Arevalo J, et al. Detection and species identification of Leishmania DNA from filter paper lesion impressions for patients with American cutaneous leishmaniasis. Clinical Infectious Diseases. 2010; 50(1):e1–e6. https://doi.org/10.1086/648730 PMID: 19947858
- 35. de Morais RCS, de Melo MGN, de Goes TC, e Silva RP, de Morais RF, de Oliveira Guerra JA, et al. Duplex qPCR for Leishmania species identification using lesion imprint on filter paper. Experimental parasitology. 2020; 219:108019. https://doi.org/10.1016/j.exppara.2020.108019 PMID: 33053400
- Mendez-Bejarano CP, Perez-Rico JJ, Rodriguez-Angarita O, Velandia-Rodriguez DA, Romero-Barbosa YA, Alvarado-Flechas MT, Valbuena-Pinzon T. the imprint as a non-invasive method of sampling for the molecular diagnosis of cutaneous leishmaniasis in Colombia's Military population. Revista Med. 2018; 26(2):15–21.
- 37. Thomaz C, de Mello CX, Espindola OdM, Shubach AdO, Quintella LP, de Oliveira RVC, et al. Comparison of parasite load by qPCR and histopathological changes of inner and outer edge of ulcerated cutaneous lesions of cutaneous leishmaniasis. PloS one. 2021; 16(1):e0243978. https://doi.org/10.1371/journal.pone.0243978 PMID: 33476320
- Simon N, Shallat J, Williams Wietzikoski C, Harrington WE. Optimization of Chelex 100 resin-based extraction of genomic DNA from dried blood spots. Biology Methods and Protocols. 2020; 5(1): bpaa009. https://doi.org/10.1093/biomethods/bpaa009 PMID: 32665976
- Adams ER, Gomez MA, Scheske L, Rios R, Marquez R, Cossio A, et al. Sensitive diagnosis of cutaneous leishmaniasis by lesion swab sampling coupled to qPCR. Parasitology. 2014; 141(14):1891–7. https://doi.org/10.1017/S0031182014001280 PMID: 25111885
- 40. Kato H, Gomez EA, Martini-Robles L, Muzzio J, Velez L, Calvopiña M, et al. Geographic Distribution of Leishmania Species in Ecuador Based on the Cytochrome B Gene Sequence Analysis. PLoS neglected

tropical diseases. 2016; 10(7):e0004844. https://doi.org/10.1371/journal.pntd.0004844 PMID: 27410039

- Calvopina M, Armijos RX, Hashiguchi Y. Epidemiology of leishmaniasis in Ecuador: current status of knowledge—a review. Memorias do Instituto Oswaldo Cruz. 2004; 99(7):663–72. https://doi.org/10. 1590/s0074-02762004000700001 PMID: 15654419
- 42. Amunárriz M. Estudios sobre patologías tropicales en la Amazonia ecuatoriana: Cicame; 1991.
- Boggild AK, Ramos AP, Valencia BM, Veland N, Calderon F, Arevalo J, et al. Diagnostic performance of filter paper lesion impression PCR for secondarily infected ulcers and nonulcerative lesions caused by cutaneous leishmaniasis. Journal of clinical microbiology. 2011; 49(3):1097–100. <u>https://doi.org/10. 1128/JCM.02457-10 PMID: 21177908</u>
- Romero GA, de Farias Guerra MV, Paes MG, Macêdo VdO. Comparison of cutaneous leishmaniasis due to Leishmania (Viannia) braziliensis and L.(V.) guyanensis in Brazil: clinical findings and diagnostic approach. Clinical Infectious Diseases. 2001; 32(9):1304–12. <u>https://doi.org/10.1086/319990</u> PMID: 11303265
- Linnet K, Bossuyt PM, Moons KG, Reitsma JB. Quantifying the accuracy of a diagnostic test or marker. Clinical chemistry. 2012; 58(9):1292–301. <u>https://doi.org/10.1373/clinchem.2012.182543</u> PMID: 22829313
- 46. Aronson N, Herwaldt BL, Libman M, Pearson R, Lopez-Velez R, Weina P, et al. Diagnosis and treatment of leishmaniasis: clinical practice guidelines by the Infectious Diseases Society of America (IDSA) and the American Society of Tropical Medicine and Hygiene (ASTMH). Clinical infectious diseases. 2016; 63(12):e202–e64. https://doi.org/10.1093/cid/ciw670 PMID: 27941151
- **47.** Urutseg. Ecuador physical map 2014 <u>https://en.m.wikipedia.org/wiki/File:Ecuador\_relief\_location\_map.</u> svg. Accessed: 20 January 2023.